Author's response to reviews

Title: Health problems and disability in long-term sickness absence: ICF coding of medical certificates

Authors:

Roland Morgell (torkel.morgell@sll.se)
Lars G Backlund (lars.g.backlund@ki.se)
Britt Arrelöv (britt.arrelov@sll.se)
Lars-Erik Strender (lars-erik.strender@ki.se)
Gunnar H Nilsson (gunnar.nilsson@ki.se)

Version: 5 Date: 10 November 2011

Author's response to reviews:

The title has been changed as remarked.
The additional files have been removed.
The vertical lines in the Tables have been removed.